Author's response to reviews

Title: Association between LRP5 polymorphism and bone mineral density: a Bayesian meta-analysis

Authors:

Bich N.H. Tran (b.tran@garvan.org.au)
Nguyen D. Nguyen (n.nguyen@garvan.org.au)
John A. Eisman (j.eisman@garvan.org.au)
Tuan V. Nguyen (t.nguyen@unsw.garvan.org.au)

Version: 6 Date: 13 June 2008

Author's response to reviews:

Dear Dr Manning,

We have finished the proofing stage. We hope that the manuscript is now suitable for publication in the BMC Medical Genetics.

Yours sincerely,

On behalf of the authors

Tuan V Nguyen